Author's response to reviews

Title: Repeal of the Pennsylvania motorcycle helmet law: reflections on the ethical and political dynamics of public health reform

Authors:

Robert A. Cherry (rcherry@psu.edu)

Version: 3 Date: 15 April 2010

Author's response to reviews: see over
April 15, 2010

Robert. A. Cherry, MD, MS, FACS  
Associate Professor of Surgery and Public Health Sciences  
Associate Chief Quality Officer  
Penn State Milton S. Hershey Medical Center  
500 University Drive  
Hershey, Pennsylvania  17033  
Telephone: 717-531-6241  
Fax: 717-531-0109  
E-mail: rcherry@psu.edu

BMC Series Editorial Production Team  
*BMC Public Health*

**RE: MS 3224359113188945, entitled "Repeal of the Pennsylvania motorcycle helmet law: reflections on the ethical and political dynamics of public health reform"**

Dear BMC Series Editorial Production Team,

I have carefully read the e-mail and have made the requested changes to the manuscript.

Respectfully submitted,

Robert A. Cherry, MD, MS, FACS
Formatting changes that were made to the manuscript:

Abstract: abstracts should not cite references, nor refer to figures or tables
   The reference was removed and all references were re-numbered.

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings:
Background, Discussion, Summary
Please remember to also update the abstract details on the submission page.
   The abstract headings were corrected as noted.

Debate: please remove this text from above the title
   Debate was removed as requested.

Title: please use sentence case in the title, both in the manuscript file and the submission system.
   Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, ?The Title of My Prestigious Publication: A Scientific Study.? should be changed to ?The title of my prestigious publication: a scientific study?
   The title was corrected as requested.

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system
   All qualifications of the author was removed on the title page.

Email address: please include email addresses for all the authors on the title page, using the following format:
   AB: abcd@institution.ac.uk
   EF: efgb@generic.co.uk
   IJ: IJKL@corporation.com
   Author’s e-mail address was added.

DEBATE: please remove this test from above the main body of the manuscript
   DEBATE was removed from the body of the manuscript.

Tracking: please remove all tracking and strikethrough from the manuscript
   All tracking, bolding, and strikethroughs were removed.

Abbreviations: Please format these in to a paragraph.
   The abbreviations were arranged in paragraph form.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
   The manuscript was inspected for typographical errors.

Thank you